Author's response to reviews

Title: Comparison of the efficacy and safety of two starting dosages of prednisolone in early active rheumatoid arthritis (CORRA): study protocol for an investigator-initiated, randomized, double-blind, placebo-controlled trial

Authors:

Ulrike S Trampisch (ulrike-trampisch@amib.ruhr-uni-bochum.de)
Dietmar Krause (krause@amib.ruhr-uni-bochum.de)
Hans J Trampisch (trampisch@amib.ruhr-uni-bochum.de)
Renate Klaassen-Mielke (klaassen-mielke.de@amib.ruhr-uni-bochum.de)
Xenofon Baraliakos (baraliakos@me.com)
Rheumanetz Westfalen Lippe (krause@amib.ruhr-uni-bochum.de)
Jürgen Braun (j.braun@rheumazentrum-ruhrgebiet.de)

Version: 2
Date: 11 March 2014

Author's response to reviews:

Dear Jessie McGavin,

According to your email we have made the remarked details.

We will enclose the original funding and ethics documentation. Unfortunately, these documents are in German. I hope it will be sufficient. If not, please contact us again.

Thank you.